Author's response to reviews

Title: Job stress and the utilization of health care services

Authors:

Sunday Azagba (s_azagba@live.concordia.ca)
Mesbah F Sharaf (m_shara@live.concordia.ca)

Version: 2 Date: 6 June 2011

Author's response to reviews:

The attached supplementary files in the previous submission was in word 2007 version. In this revised version, we are attaching it as a 2003 word document.